Author's response to reviews

Title: MAK-4 and -5 supplemented diet inhibits liver carcinogenesis in mice

Authors:

Marialetizia Penza (marialetziapenza@inwind.it)
Claudia Montani (claudia.montani@hotmail.it)
Marija Jeremic (mjericmic@hotmail.com)
Giovanna Mazzoleni (mazzoleni@med.unibs.it)
Wendy W.L. Hsiao (bowhsiao@hkbu.edu.hk)
Maurizio Marra (m.marra@inrca.it)
Hari Sharma (sharma-1@medcetr.osu.edu)
Diego Di Lorenzo (dilorenzodiego@yahoo.it)

Version: 3 Date: 30 May 2007

Author's response to reviews:

To the Editor of
BMC Complementary and Alternative Medicine

Dear Editor,

Please find enclosed manuscript of:
MAK-4 and -5 supplemented diet inhibits liver carcinogenesis in mice.
By
Marialetizia Penza, Claudia Montani, Marija Jeremic, Giovanna Mazzoleni, W.L. Wendy Hsiao, Maurizio Marra, Hari Sharma and Diego Di Lorenzo

The work was reviewed according to the formatting changes requested.

* Unnecessary capitalizations were removed from subheadings
* "Conclusion" was converted in "conclusions"
* Figure numbers were removed from the image files
* Figures were cropped as closely as possible
* Website was removed from the text and cited only in the main reference list (reference 43)

Best regards,

Diego Di Lorenzo
Laboratory of Biotechnology
Ospedale Civile di Brescia
P.le Spedali Civili 1
I-25123 Brescia (Italy)
Phone 0039 030 3995408-7-6
FAX 0039 030 307251
Mobile Phone 3355923653
e-mail: dilorenzodiego@yahoo.it
websites: www.spedalicivili.brescia.it/
http://www.med.unibs.it